Author's response to reviews

Title: Spontaneous closure of patent ductus arteriosus after an episode of Kawasaki disease: a case report

Authors:

Ming-Chih Lin (mingclin@gmail.com)
Yun-Ching Fu (ivanfu@vghtc.gov.tw)
Sheng-Ling Jan (sljan@vghtc.gov.tw)

Version: 2 Date: 11 September 2011

Author's response to reviews:

1. Re: Please include the ethnicity of the patient in the abstract
Ans: The ethnicity has been added.

2. Please include the ethnicity of the patient in the case presentation section of the manuscript.
Ans: The ethnicity has been added.

3. Re: As the patient is a minor, please could you confirm that you have obtained written informed consent from the patient’s legal guardian for publication of the case. Your consent statement should be modified accordingly.
Ans: The section of informed consent has been revised.

4. Re: Please restructure the authors contribution section. We suggest the following kind of format (please use initials to refer to each author’s contribution): FC analyzed and interpreted the patient data regarding the hematological disease and the transplant. RH performed the histological examination of the kidney, and was a major contributor in writing the manuscript. All authors read and approved the final manuscript.
Ans: The section of informed consent has been revised.

5. Re: Please remove any text from figure 1 which may reveal the patient’s identity (e.g. patient’s number, patient’s name, dates, etc)
Ans: The text on the picture has been further scrambled.